Author’s response to reviews

Title: Patient Centered Psoriatic Arthritis (PsA) Activity Assessment by the Stockerau Activity Score for Psoriatic Arthritis (SASPA)

Authors:

Burkhard F Leeb (leeb.rheuma@aon.at)
Pia M Haindl (piahai@gmx.net)
Hans-Peter Brezinschek (hans-peter.brezinsek@medunigraz.at)
Harsono TH Mai (the-hien-harsono.mai@stockerau.lknoe.at)
Christoph Deutsch (christoph.deutsch@stockerau.lknoe.at)
Bernhard Rintelen (b.rintelen@gmx.at)

Version: 2
Date: 1 August 2014

Author’s response to reviews:

To the Editor:
The title page was reformatted according to the prerequisites and contains now the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
The article was divided into
Background
Methods
Results
Discussion, and
Conclusions
A list of abbreviations was included
Line and page numbers were included.
A sentence “The authors declare that they have no competing interests.” was included
The Authors’ Contributions section was also included and placed after the Competing Interests section.
I hope that the manuscript is now acceptable for evaluation,
Yours Sincerely,
Burkhard F. Leeb, MD, PhD